Author’s response to reviews

Title: Socioeconomic inequalities in suicide mortality before and after the economic recession in Spain

Authors:

Carme Borrell (cborrell.aspb@gmail.com)
Marc Marí Dell’Olmo (mmari@aspb.cat)
Mecè Gotsens (mgotsens@aspb.cat)
Montse Calvo (mcalvo@ej-gv.es)
Maica Rodríguez-Sanz (mrodri@aspb.cat)
Xavier Bartoll (xbartoll@aspb.cat)
Santi Esnaola (sesnaola@euskadi.eus)

Version: 1 Date: 21 Nov 2016

Author’s response to reviews:

We have added next paragraph:

Use of third party data

The Agència de Salut Pública de Barcelona (ASPB), as health authority for the city of Barcelona has access to the data used in this study: census, mortality and forensic registry data. These sources of information form part of the routine sources used in the health information system of Barcelona of ASPB.